Author's response to reviews

Title: An integrated analysis of genes and pathways exhibiting metabolic differences between estrogen receptor positive breast cancer cells

Authors:

Soma SM Mandal (mandals0@cc.umanitoba.ca)
James JRD Davie (davie@ms.umanitoba.ca)

Version: 6 Date: 14 September 2007

Author's response to reviews: see over
Re.: MS: 6534271741313271

We are hereby submitting the manuscript with the formatting changes as suggested.

As I will be away for some time now, I request you to direct all correspondence to Dr. Soma Mandal (mandals0@cc.umanitoba.ca) in the interim.

Hope to hear back from you soon.

Sincerely yours,

James Davie, Ph.D.
Professor, Department of Biochemistry and Medical Genetics
Canada Research Chair in Chromatin Dynamics
Director and Margaret A. Sellers Chair, MICB
Provincial Director, Research, CancerCare Manitoba
675 McDermot Avenue
Winnipeg, Manitoba R3E 0V9 Canada
Tel: (204)787-2391
Fax: (204)787-2190
E-mail: davie@ms.umanitoba.ca